Author's response to reviews

Title: A retrospective review of telehealth services for children referred to a paediatric nephrologist

Authors:

Peter Trnka (peter.trnka@health.qld.gov.au)
Megan M White (megan.white@uq.edu.au)
William D Renton (wdrenton@gmail.com)
Steven J McTaggart (steven.mctaggart@health.qld.gov.au)
John R Burke (jburke@gil.com.au)
Anthony C Smith (asmith@uq.edu.au)

Version: 4 Date: 24 July 2015

Author's response to reviews: see over
July 24, 2015

Dr. Hayley Henderson, PhD
Executive Editor
BioMed Central
Floor 6, 136 Gray’s Inn Road
London
WC1X 8HB
United Kingdom

Dear Dr. Henderson,

Thank you for accepting our manuscript “A retrospective review of telehealth services for children referred to a paediatric nephrologist” for publication in BMC Nephrology journal. As requested by the editors, we have made the following changes to the manuscript:

1. Full name of the local ethics committee has been added to the first paragraph on page 6
2. The tables have been moved to after the reference section
3. The map depicted in figure 1 was created for this project by the authors and was not taken from another source

We hope that the manuscript can now be published in your journal.

Sincerely,

[Signature]

Peter Trnka, FRACP, MSc
Pediatric Nephrologist
Department of Paediatrics and Child Health
The University of Queensland
Brisbane
Australia